Author's response to reviews

Title: An analysis of trends and determinants of health insurance and healthcare utilisation in the Russian population between 2000 and 2004: the 'inverse care law' in action

Authors:

Francesca Perlman (francesca.perlman@lshtm.ac.uk)
Dina Balabanova (dina.balabanova@lshtm.ac.uk)
Martin McKee (martin.mckee@lshtm.ac.uk)

Version: 4 Date: 15 April 2009

Author's response to reviews:

I am very sorry for submitting the incorrect version of the manuscript. Here is the correct one, with the changes made as listed above.